Author's response to reviews

Title: Myasthenia gravis and pregnancy: Clinical implications and neonatal Outcome

Authors:

Jose Francisco Tellez-Zenteno (jftellez@quetzal.innsz.mx)
Lizbeth Hernandez-Ronquillo (liztellez33@hotmail.com)
Vicente Salinas (vicente.salinas@latinmail.com)
Guillermo Garcia Ramos (garcira@quetzal.innsz.mx)
Bruno Estanol (bestanol@hotmail.com)
da Silva Orlando (odasilva@julian.uwo.ca)

Version: 7 Date: 14 July 2004

Author's response to reviews: see over
We added the information related with the status of acetylcholine receptors.

The patients became pregnant at a mean of 2 years post-thymectomy. In our centre we prefer to do the thymectomy first and then when the patient is stable we recommend the pregnancy. This is not a generalized protocol in many centers but the majority of our patients were stable before the pregnancy. Obviously it does not mean that the prognosis is going to be good during the pregnancy but the majority of patients were in good conditions and they were not taking many medications. We included this information in patients and methods.

In our point of view, we do not want to separate the clinical severity and the change in drug dose. Our system was based in the combination of both factors and finally we have four well defined groups to describe the prognosis.

We changed the institutional address

We changed the information related with the length of pregnancy,

The article was reviewed for an expert in the English language.

We changed the mistakes in the references

We deleted the table 3, the information is included in the discussion